
-----------------------------------------------------------------------------
          Mauna Loa Solar Observatory Observer’s Log 
-----------------------------------------------------------------------------
       Fri Jun 19 16:19:50 GMT 2015
Year: 15  Doy: 170
Observer: berkey
WEATHER COMMENT: berkey: Fri Jun 19 16:22:46 GMT 2015
Temp: 46.4f, Humidity: 22%, Pressure: 28.621in, Wind: 10mph from the south, Skies: thin cirrus
___end___
WEATHER COMMENT: berkey: Fri Jun 19 16:22:58 GMT 2015
Temp: 46.5f, Humidity: 23%, Pressure: 28.625in, Wind: 13mph from 360degs, Skies:
___end___
WEATHER COMMENT: berkey: Fri Jun 19 16:23:01 GMT 2015
Temp: 46.4f, Humidity: 22%, Pressure: 28.621in, Wind: 10mph from the south, Skies: thin cirrus
___end___
GENERAL COMMENT BY berkey: Fri Jun 19 16:24:16 GMT 2015
Day 2 of kcor new camera plate install
___end___
KCOR COMMENT BY berkey: Fri Jun 19 20:04:36 GMT 2015
Optical alignment in tip/tilt is done.  Taking some distoration masks flats.
___end___
Fri Jun 19 20:31:16 GMT 2015 COMP Start Patrol
KCOR COMMENT BY berkey: Fri Jun 19 21:01:54 GMT 2015
Yesterday we reverted to the old cameralink cable on the reflected beam.  Today I couldn’t get socketcam running got lots 
got board acquisition errors on Brd1.  Reverting to the old cable on this camera fixed this.

Taking distortion mask data on kcor from 20:59:24 to 21:00:57
___end___
CoMP COMMENT BY berkey: Fri Jun 19 21:17:59 GMT 2015
I didn’t noticed it for a bit, but we took a bit of comp data on the inside of the dome shutter.  So we should see a bit o
f junk over the last 15 minutes
___end___
KCOR COMMENT BY berkey: Fri Jun 19 21:55:31 GMT 2015
I tried to tidy up the cameralink cables and now I am having socketcam problems again.  :(  

I tried rebooting and re-seating the cables a couple times no success.

Giving up for the day.
___end___
Fri Jun 19 21:56:28 GMT 2015 COMP End Patrol


